Author's response to reviews

Title: Evaluation of antiarthritic activity of Strychnos potatorum Linn seeds in Freund's adjuvant induced arthritic rat model

Authors:

Sanmugapriya Ekambaram (sanmug77@gmail.com)
Venkataraman Subramanian (svee1944@rediffmail.com)

Version: 7 Date: 15 July 2010

Author's response to reviews: see over
Reviewer's report
Title: Evaluation of antiarthritic activity of Strychnos potatorum Linn seeds in Freund's adjuvant induced arthritic rat model
Version: 6 Date: 22 June 2010
Reviewer: Jose Luis Rios
Reviewer's report:
The paper was clearly improved and some minor mistakes should be modified.
page 6, lines 6 and 20 (Wistar albino rats (140 ± 20 gms)change gms for g (...20 g)
pge 7, change x (letter) by x (symbol) in x100

All the corrections mentioned by the reviewer have been made.

Reviewer's report
Title: Evaluation of antiarthritic activity of Strychnos potatorum Linn seeds in Freund's adjuvant induced arthritic rat model
Version: 6 Date: 13 June 2010
Reviewer: Kamal Moudgil
Reviewer's report:
Please check if the radiographs meet the quality for publication

The quality of the radiograph has been improved and uploaded

- Please provide under the methods section ?page 7- a better description of the samples employed, and how they were homogenized.

The preparation of the extract is mentioned under a separate heading. Anyhow it is now mentioned in the specified part again as asked by the reviewer.

- Please provide better quality images for figures 2 and 3.

The quality of the images have been improved and attached.

- Please provide under the results and discussions section a better description and discussion of the data reported in the figures.

The histopathology and radiology of the joints are now discussed in detail in the results and discussion part.